Author's response to reviews

Title: OPTimizing IMaging in suspected APpendicitis (OPTIMAP-study): A multicenter diagnostic accuracy study of MRI in patients with suspected acute appendicitis. Study Protocol

Authors:

Marjolein M.N. Leeuwenburgh (m.m.leeuwenburgh@amc.uva.nl)
Wytze Laméris (w.lameris@amc.uva.nl)
Adrienne van Randen (a.vanranden@amc.uva.nl)
Patrick M.M. Bossuyt (p.m.bossuyt@amc.uva.nl)
Marja A. Boermeester (m.a.boermeester@amc.uva.nl)
Jaap Stoker (j.stoker@amc.uva.nl)
OPTIMAP study group (OPTIMAP) (m.m.leeuwenburgh@amc.uva.nl)

Version: 4 Date: 15 October 2010

Author's response to reviews:

Dear BMC Series Editorial Production Team,

Thank you for accepting our manuscript for publication in BMC Emergency Medicine. Most of the formatting changes you requested have been made, and resubmitted. Can you please adjust the final layout of the figure 1 (minimise white space around the image)?

Major revisions:
- We included the name of the OPTIMAP study group within the list of authors, by adding the name to the 'first name' box; and the bracketed abbreviation to the 'last name' box in your submission system.
- The names of the members of the group are found under 'acknowledgements'

Minor revisions:
- We removed the text 'Manuscript' from above the main body of the text.
- Full citation details are now given for references 11 and 21.
- The references are now placed before the full stop as you requested.
- We checked our manuscript for any typographical errors and didn't find any.

Best regards,

Marjolein Leeuwenburgh, MD

Research fellow OPTIMAP study
Department of Surgery/Radiology